Title: Quality Assessment of Reporting of Randomization, Allocation Concealment, and Blinding in Traditional Chinese Medicine RCTs: A Review of 3159 RCTs identified from 260 Systematic Reviews

Authors:

He Jia (hejia1986@hotmail.com)
Du Liang (duliang0606@vip.sina.com)
Liu Guanjian (ceuliu@hotmail.com)
Fu Jin (jfu1126@163.com)
He Xiangyu (447415665@qq.com)
Yu Jiayun (389585399@qq.com)
Shang Lili (282793228@qq.com)

Version: 7 Date: 4 April 2011

Author's response to reviews: see over
Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is ‘First name Initial Last name’ (e.g. Joe F Bloggs)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: We have checked the author list, affiliations and typography conforms to journal style.